

PEER REVIEW HISTORY

BMJ Open publishes all reviews undertaken for accepted manuscripts. Reviewers are asked to complete a checklist review form (<http://bmjopen.bmj.com/site/about/resources/checklist.pdf>) and are provided with free text boxes to elaborate on their assessment. These free text comments are reproduced below. Some articles will have been accepted based in part or entirely on reviews undertaken for other BMJ Group journals. These will be reproduced where possible.

ARTICLE DETAILS

TITLE (PROVISIONAL)	CAFÉ - a multi-component audit and feedback intervention to improve implementation of healthy food policy in primary school canteens: protocol of a randomised controlled trial
AUTHORS	Williams, Christophe ⁰⁰⁷² ; Nathan, Nicole; Delaney, Tessa; Yoong, Sze Lin; Wiggers, John; Preece, Sarah; Lubans, Nicole; Sutherland, Rachel; Pinfold, Jessica; Smith, Kay; Small, Tameka; Reilly, Kathryn; Butler, Peter; Wyse, Rebecca; Wolfenden, Luke

-This article was not externally reviewed at BMJ Open. Protocols that have been independently assessed prior to submission to BMJ Open are usually fast-tracked to publication on the grounds that further substantial changes will not be possible. This independent assessment will usually be external, independent review for both a substantial grant award from a non commercial or government funder and ethics approval.